Author's response to reviews

Title: Enhancing provision of written medicine information in Australia: pharmacist, general practitioner and consumer perceptions of the barriers and facilitators

Authors:

Kim K Hamrosi (kim.hamrosi@sydney.edu.au)
David K Raynor (D.K.Raynor@leeds.ac.uk)
Parisa Aslani (parisa.aslani@sydney.edu.au)

Version: 4 Date: 31 March 2014

Author's response to reviews: see over
Date: 31st March, 2014

Prof David Preen
BMC Health Services Research
BioMed Central
236 Gray's Inn Road
London WC1X 8HB United Kingdom

Dear Professor Preen,

Thank you for your email dated 19th March 2014 regarding our manuscript “Enhancing provision of written medicine information in Australia: pharmacist, general practitioner and consumer perceptions of the barriers and facilitators ” for publication into BMC Health Services Research.

In your email you asked that we confirm whether we have permission to reproduce these questionnaires and for them to featured in a BioMed Central publication. We have sought clarification from the funding bodies regarding permission to reprint these in BMC Health Services Research. Unfortunately, the process has been rather slow to progress and the terms of the contracts between the University and funding bodies as to reproduction of the survey materials and copyright not entirely clear. Our article does however reference the surveys that are available in the larger project report, which the above study formed part of. At this stage, until we are able to receive a definitive answer on the reproduction of the surveys elsewhere, we are unable to give permission for them to be reprinted with BMC Health Services Research. We apologise if this has caused any inconvenience.

Thank you again for accepting our article for publication. Should you require any further information or clarification regarding this or pertaining to our article please do not hesitate to contact me.

Sincerely,

Ms. Kim Hamrosi
Corresponding author

(on behalf of A/Prof Aslani and Prof D.K. Raynor)